**Author's response to reviews**

**Title:** Mentorship needs at academic institutions in resource-limited settings: A Survey at Makerere University College of Health Sciences

**Authors:**

Damalie Nakanjako (drdamalie@yahoo.com)
Pauline Byakika-Kibwika (pbyakika@gmail.com)
Kenneth Kintu (kkintu@mujhu.org)
Jim Aizire (jaizire@mujhu.org)
Fred Nakwagala (nakwagala@yahoo.com)
Simon Luzige (drlsmluz@hotmail.com)
Charles Namisi (namisipc@yahoo.co.uk)
Harriet Mayanja-Kizza (hmk@chs.mak.ac.ug)
Moses R. Kamya (mkamya@infocom.co.ug)

**Version:** 8 **Date:** 25 July 2011

**Author's response to reviews:** see over
The authors proof read and made corrections on typos and spelling